////
Automatically generated by PainlessDocGenerator. Do not edit.
Rebuild by running `gradle generatePainlessApi`.
////

[[painless-api-reference-NavigableSet]]++NavigableSet++::
* ++[[painless-api-reference-NavigableSet-ceiling-1]]def link:{java8-javadoc}/java/util/NavigableSet.html#ceiling%2Djava.lang.Object%2D[ceiling](def)++ (link:{java9-javadoc}/java/util/NavigableSet.html#ceiling%2Djava.lang.Object%2D[java 9])
* ++[[painless-api-reference-NavigableSet-descendingIterator-0]]<<painless-api-reference-Iterator,Iterator>> link:{java8-javadoc}/java/util/NavigableSet.html#descendingIterator%2D%2D[descendingIterator]()++ (link:{java9-javadoc}/java/util/NavigableSet.html#descendingIterator%2D%2D[java 9])
* ++[[painless-api-reference-NavigableSet-descendingSet-0]]<<painless-api-reference-NavigableSet,NavigableSet>> link:{java8-javadoc}/java/util/NavigableSet.html#descendingSet%2D%2D[descendingSet]()++ (link:{java9-javadoc}/java/util/NavigableSet.html#descendingSet%2D%2D[java 9])
* ++[[painless-api-reference-NavigableSet-floor-1]]def link:{java8-javadoc}/java/util/NavigableSet.html#floor%2Djava.lang.Object%2D[floor](def)++ (link:{java9-javadoc}/java/util/NavigableSet.html#floor%2Djava.lang.Object%2D[java 9])
* ++[[painless-api-reference-NavigableSet-headSet-2]]<<painless-api-reference-NavigableSet,NavigableSet>> link:{java8-javadoc}/java/util/NavigableSet.html#headSet%2Djava.lang.Object%2Dboolean%2D[headSet](def, boolean)++ (link:{java9-javadoc}/java/util/NavigableSet.html#headSet%2Djava.lang.Object%2Dboolean%2D[java 9])
* ++[[painless-api-reference-NavigableSet-higher-1]]def link:{java8-javadoc}/java/util/NavigableSet.html#higher%2Djava.lang.Object%2D[higher](def)++ (link:{java9-javadoc}/java/util/NavigableSet.html#higher%2Djava.lang.Object%2D[java 9])
* ++[[painless-api-reference-NavigableSet-lower-1]]def link:{java8-javadoc}/java/util/NavigableSet.html#lower%2Djava.lang.Object%2D[lower](def)++ (link:{java9-javadoc}/java/util/NavigableSet.html#lower%2Djava.lang.Object%2D[java 9])
* ++[[painless-api-reference-NavigableSet-pollFirst-0]]def link:{java8-javadoc}/java/util/NavigableSet.html#pollFirst%2D%2D[pollFirst]()++ (link:{java9-javadoc}/java/util/NavigableSet.html#pollFirst%2D%2D[java 9])
* ++[[painless-api-reference-NavigableSet-pollLast-0]]def link:{java8-javadoc}/java/util/NavigableSet.html#pollLast%2D%2D[pollLast]()++ (link:{java9-javadoc}/java/util/NavigableSet.html#pollLast%2D%2D[java 9])
* ++[[painless-api-reference-NavigableSet-subSet-4]]<<painless-api-reference-NavigableSet,NavigableSet>> link:{java8-javadoc}/java/util/NavigableSet.html#subSet%2Djava.lang.Object%2Dboolean%2Djava.lang.Object%2Dboolean%2D[subSet](def, boolean, def, boolean)++ (link:{java9-javadoc}/java/util/NavigableSet.html#subSet%2Djava.lang.Object%2Dboolean%2Djava.lang.Object%2Dboolean%2D[java 9])
* ++[[painless-api-reference-NavigableSet-tailSet-2]]<<painless-api-reference-NavigableSet,NavigableSet>> link:{java8-javadoc}/java/util/NavigableSet.html#tailSet%2Djava.lang.Object%2Dboolean%2D[tailSet](def, boolean)++ (link:{java9-javadoc}/java/util/NavigableSet.html#tailSet%2Djava.lang.Object%2Dboolean%2D[java 9])
* Inherits methods from ++<<painless-api-reference-Collection,Collection>>++, ++<<painless-api-reference-Iterable,Iterable>>++, ++<<painless-api-reference-Object,Object>>++, ++<<painless-api-reference-Set,Set>>++, ++<<painless-api-reference-SortedSet,SortedSet>>++
